Author's response to reviews

Title: Assessment of an enhanced program for depression management in primary care: a cluster randomized controlled trial. The INDI project (Interventions for Depression Improvement).

Authors:

Enric Aragones (earagones.tarte.ics@gencat.net)
Antonia Caballero (antoniacaballero19@hotmail.com)
Josep Ll. Pinol (gcortacans@mixmail.com)
German Lopez (jlpinyl.tarte.ics@gencat.net)
Waleska Badia (wbadia.tarte.ics@gencat.net)
Josep M. Hernandez (jmhernan@comt.org)
Pilar Casaus (casausp@peremata.com)
Silvia Folch (sfolch.tarte.ics@gencat.net)
Josep Basora (jbasora.tarte.ics@gencat.net)
Antonio Labad (labada@peremata.com)

Version: 3 Date: 14 September 2007

Author's response to reviews:

Dear editor,

We have done the changes to the format of our manuscript for that you have asked us. The Abstract does not include the paragraph of Discussion because it is not appropriate.

Regards,

Enric Aragonès